**Author’s response to reviews**

**Title:** A multimodal and multidisciplinary program to prevent loss of mobility in patients aged over 70 years: study protocol of a multicenter cluster randomized study in primary care (the PRISME-3P study)

**Authors:**

sofia perrotin (sofia.perrotin@univ-lyon1.fr)

Thomas Gilbert (thomas.gilbert@chu-lyon.fr)

Marine Dupuis (marine.dupuis01@chu-lyon.fr)

Laurent Villeneuve (laurent.villeneuve@chu-lyon.fr)

Sylvie Bin-Dorel (sylvie.bin@chu-lyon.fr)

Amna Klich (amna.klich@chu-lyon.fr)

Laurent Letrilliart (laurent.letrilliart@univ-lyon1.fr)

Marc Bonnefoy (marc.bonnefoy@chu-lyon.fr)

**Version:** 2 **Date:** 22 Nov 2018

**Author’s response to reviews:**

1) Please include the email address of Amna Klich on the title page.

The email address has been added (P.2)

2) Please include section headings in your abstract as outlined in our Submission Guidelines (https://bmcgeriatr.biomedcentral.com/submission-guidelines/preparing-your-manuscript/study-protocol#preparing+your+manuscript). In addition, ensure that the abstract on the submission system (Editorial Manager) is also updated.

We have added "Background, method, discussion" p.2 at the abstract

3) Please remove the section heading “Conclusion” and integrate the text into the Discussion or remove if not required.

We have integrate the text into the discussion
4) Please move the Table titles to above each table and any table legends underneath the table (https://bmcgeriatr.biomedcentral.com/submission-guidelines/preparing-your-manuscript#preparing+tables).

We have added title at the table 1 and 2, and Title and legend at the table 3.